**To Reviewer #1:**

Specific Comments to Authors: This paper entitled “A nomogram for predicting the prognosis of tumor patients with sepsis after gastrointestinal surgery” is interesting and valuable. I think this manuscript is worthy for publication in World Journal of Gastrointestinal Oncology.

Reply: Thank you very much for your comments.

**To Reviewer #2:**

Specific Comments to Authors: Dear Hong-Zhi Wang, MD, Professor, this manuscript is well written but you should focus on the following comments 1. Regarding Institutional Review Board Approval Form or Document should be provided in English language. 2. In the discussion please do not repeat the results and P value should not be provide in this section. 3. Regarding the result section, I did not see any tables, figures and supplementary files, they mentioned only in the text!!!!!!!?? 4. Regarding the follow-up patients please write in detail. 5. Please provide a list of abbreviation used in the entire manuscript. 6. I think it is better to write a new paragraph in the method section regarding Surgical Technique in detail. 7. The overall quality of the manuscript is good.

Reply: 1. We have added the English language form of Institutional Review Board Approval Form or Document in the supplementary material.

2. In the discussion section, the P value was deleted.

3. We have added the tables, figures in the relevant section and supplementary material.

4. We added “and patient’s survival status was recorded” in the data collection and follow up section.

5. We have provided a list of abbreviation used in the entire manuscript in the supplementary material.

6. We have added a new paragraph in the method section regarding Surgical Technique.
7. Thank you very much.

**To Reviewer #3:**

Specific Comments to Authors: The authors discuss predictors of post-surgical infection. Predictors of postoperative infection and sepsis are considered to be very important factors for perioperative care providers. Although the data showed that these are significantly related factors, it is necessary to clarify why they are involved and to discuss them in the future.

Reply: Thank you for your comments and suggestions.